Author's response to reviews

Title: An unusual variant of Choledochal Cyst: A case report

Authors:

Javaid Sadiq (mjavaidsadiq@yahoo.com)
Biplap Nandi (bipnandi@hotmail.com)
Kokila Lakhoo (kokila.lakhoo@paediatrics.ox.ac.uk)

Version: 7 Date: 2 October 2008

Author's response to reviews: see over
Dear Colleague

Thank you for your response on Case report No. MS: 1337990131787697. We have read the reviewers comments and made appropriate changes as outlined below:

Reviewer 1:
1. The text of the case report is extended.
2. The detailed description of the postnatal scans included in the text.
3. Figure 2 is now clarified.

Reviewer 2:
The intra-operative figure (fig.1) is fully labelled as requested.

We look forward to your final decision

Yours sincerely,

Mr Javaid Sadiq
Mr Biplap Nandi
Miss Kokila Lakhoo